Author's response to reviews

Title: Risk Score to Predict Gastrointestinal Bleeding after Acute Ischemic Stroke

Authors:

Ruijun Ji (JRJChina@sina.com)
Haipeng Shen (shenhaipeng@gmail.com)
Yuesong Pan (panyu2000@163.com)
Pengliang Wang (wangpengliang@gmail.com)
Gaifen Liu (liugaifen1997@163.com)
Yilong Wang (yilong528@gmail.com)
Hao Li (li_hao71@yahoo.com)
Aneesh B Singhal (asinghal@partners.org)
Yongjun Wang (yongjunwang1962@gmail.com)

Version: 4
Date: 10 July 2014

Author's response to reviews:

1. We provide the detailed information of IRB of the participating hospital in the CNSR and CICAS network, which was labeled with yellow color in the manuscript.
2. We change Figure 1 into Table 3. Meanwhile, we rearrange all the number of table and figure in the manuscript.